sensor adjustments

A6 sonic is not installed (waiting for CSI)

A12 sonic (1123) still reporting 8000, even though everything looks fine

A17 sonic (856) same as A12

A19 sonic (672) head was bent (lower 45 degree fitting loose). Manually resetting did not bring up sonic.
A18 sonic (744) head connector was not seated properly. Fixed

1133: Rad stand oriented and leveled

~1135-1150: Swapped Soilg and Soilc motes to correct positions. Didn't install one probe into Soilc left TRH port, since it might have been blown by +12V.
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